
Daily Status - 9/12/16
9/12/16

Summary: Light winds,  5 m/s, out of the NE, clear,
                 EC150, at site #15, was down for a period yesterday (refer to blog for more information) 

Actions past 24 hours:

 installed Gill 2D at site 8
EC150 was down for short period, Oncley got it back up

To dos:

 move rad/soil Mote #1 to another site, #1 or #9

Sensor Status: 

T/RH:  ok
P: ok
2D Gill: ok
csat u,v:  ok
csat ldiag: ok
csat w, tc: ok
EC150: ok, was down for a period of time yesterday (~11:00 am to ~10:00:pm)
motes: ok
Wetness: ok
radiation: ok
Tsoil: ok
Gsoil: ok
Qsoil: dk
Cvsoil: not responding
Vbatt: ok
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